Author's response to reviews

Title: Analysis of Network Use Behavior of Clinical Medical Students in China

Authors:

Hua yang (yangh@sj-hospital.org)
CHEN Yunxiang (chenyx@sj-hospital.org)
ZHENG Liqiang (zhenglq@sj-hospital.org)
XU Xin (xux@sj-hospital.org)
CAO Xia (caox@sj-hospital.org)

Version: 4 Date: 28 January 2014

Author's response to reviews: see over
Dear editor Aldrin Ulep:

We really appreciate your help with the manuscript "Analysis of Network Use Behavior of Clinical Medical Students in China" (#7242837001157536). We have carefully considered the reviewers' comments, and edited the manuscript according to the reviewers' suggestion.

Now we resubmit the revised copy of our manuscript to you and hope the manuscript can be considered for publication in your journal.

Best regards,

Hua Yang

Director, Professor

Library of Shengjing Hospital, China Medical University

Shenyang, 110004

P.R. of China

E-mail: yangh8888@vip.sina.com or yangh@sj-hospital.org